Author's response to reviews

Title: Refusal of treatment for acute leukemia in pregnancy: a case report

Authors:

  Syheda Latifi Hoxha (drsyheda@yahoo.com)
  Vlora A Ibishi (drvlora@hotmail.com)
  Ahmet Brovina (drabrovina@yahoo.com)
  Mynevere Hoxha (minka1956@yahoo.com)
  Shefqet Lulaj (shlulaj@yahoo.com)

Version: 4 Date: 9 April 2013

Author's response to reviews: see over
Author’s response to reviews

Title: Refusal of treatment for acute leukemia in pregnancy: a case report

Authors:
  Syheda Latifi Hoxha(dr.syhedah@yahoo.com)
  Vlora Ademi Ibishi (drvlora@hotmail.com)
  Ahmet Brovina (drabrovina@yahoo.com)
  Mynevere Hoxha(minka56@yahoo.com)
  Shefqet Lulaj (shlulaj@yahoo.com)

Version: 4  Date: 9 April 2013

Author’s response to reviews: see over
Dear Dr. Michael Kidd AM
Editor-in-Chief
Journal of Medical Case Reports

Dear Journal of Medical Case Reports Editorial Team

Thank you for consideration of our manuscript for publication in Journal of Medical Case Reports.

We have revised the above manuscript according to your editor’s comments.

Editor’s comment:
They do need to work on the English language within this manuscript. There are some neologisms, like 'obstetrical' and sentences that though they can be understood are non-standard English.

As recommended: our manuscript is edited by a native English speaker with scientific expertise.

We hope that the revised manuscript is now suitable for publication in the Journal.

Sincerely yours,
Syheda Latifi Hoxha